Author's response to reviews

Title: Quality Assessment of Reporting of Randomization, Allocation Concealment, and Blinding in Traditional Chinese Medicine RCTs: A Review of 3159 RCTs identified from 260 Systematic Reviews

Authors:

He Jia (hejia1986@hotmail.com)
Du Liang (duliang0606@vip.sina.com)
Liu Guanjian (ceuli@hotmail.com)
Fu Jin (jfu1126@163.com)
He Xiangyu (447415665@qq.com)
Yu Jiayun (389585399@qq.com)
Shang Lili (282793228@qq.com)

Version: 6 Date: 28 March 2011

Author's response to reviews: see over
MS: 1374542046677894
Methodology
Quality Assessment of Reporting of Randomization, Allocation Concealment, and Blinding in Traditional Chinese Medicine RCTs: A Review of 3159 RCTs identified from 260 Systematic Reviews
He Jia, Du Liang, Liu Guanjian, Fu Jin, He Xiangyu, Yu Jiayun and Shang Lili
Trials

Dear Liu,

Thank you for submitting your revised manuscript. I am delighted to inform you that your manuscript is potentially ready to be accepted. However, before formal acceptance, it is important that you ensure your manuscript conforms to the journal style by addressing the matters listed below:

- Please could you remove the tables and figures from the main text of the manuscript. The tables should be included at the end of the manuscript after the reference list, and the figures should be uploaded as additional figure files.

Response: We have removed the tables and figures from the main text of the manuscript. The tables were been included at the end of the manuscript after the reference list, and the figures were been uploaded as additional figure files.